Author's response to reviews

Title: Standardized ultrasound evaluation of carotid stenosis for clinical trials: University of Washington Ultrasound Reading Center

Authors:

Kirk W Beach (kwbeach@uw.edu)
Robert O Bergelin (roberg@uw.edu)
Daniel F Leotta (leotta@uw.edu)
Jean F Primozich (jeanprimozich@comcast.net)
P Max Sevareid (mountainmaximus@yahoo.com)
Edward T Stutzman (estutz@uw.edu)
R Eugene Zierler (gzierler@uw.edu)

Version: 5 Date: 4 September 2010

Author's response to reviews: see over
Cardiovascular Ultrasound

Re: Paper submission:
Standardized Ultrasound Evaluation of Carotid Stenosis for Clinical Trials:
University of Washington Ultrasound Reading Center
Journal: Cardiovascular Ultrasound
MS ID : 1875487961434908

September 4, 2010

Dear Cardiovascular Ultrasound Editors

I’ve spoken to David Botha about these questions. He will make the appropriate revisions in the text.

Major revisions (we require the author to make these changes)
REQUEST
Figures: at 100% the text of Figure 2 is not readable.
We recommend you upload a higher resolution version of the figure.
RESPONSE
A new figure has been generated and uploaded

REQUEST
Figures: at 100% the text of Figure 8 is not readable.
We recommend you upload a higher resolution version of the figure.
RESPONSE
A new figure has been generated and uploaded

REQUEST
Table citation: Page 8 currently includes a citation for 'Table 3'.
If this is referring to a table from another article, please spell out the table number (i.e. Table three).
RESPONSE
This is a direct quote. I think that changing anything within a quote is not proper.
Check with David Botha

REQUEST
Structure: Please place the author affiliations below the author list. The corresponding author should appear below the affiliation.
RESPONSE
All authors are affiliated with the DE Strandness Vascular Laboratory
David Botha has agreed to make that change

REQUEST
Abbreviations: Please include the 'abbreviation prefixes' under the 'abbreviations' heading, rather than separately.
RESPONSE
David Botha has agreed to make that change

Sincerely

Kirk W. Beach, PhD, MD
Emeritus Research Professor,
Bioengineering and Surgery

Contact 020 3192 2014

Thank you David

Kirk